**Author's response to reviews**

**Title:** Clinical implications of thymidylate synthetase, dihydropyrimidine dehydrogenase and orotate phosphoribosyl transferase activity levels in colorectal carcinoma following radical resection and administration of adjuvant 5-FU chemotherapy

**Authors:**

Masashi Ishikawa (masa1192@tokushima-med.jrc.or.jp)  
Toshihiro Ichimori (ichimori@viola.ocn.ne.jp)  
Akihiro Sakata (asakata@tokushima-med.jrc.or.jp)  
Takayuki Miyauchi (t-miyauchi@tph.gr.jp)  
Yutaka Kashiwagi (masa1192@tokushima-med.jrc.or.jp)

**Version:** 2  
**Date:** 18 June 2007

**Author's response to reviews:**

1) We would be happy to have our manuscript transferred to BMC Cance from BMC Gastroenterology.

2) We declare that we have no competing interests.

3) Co authors; TI and AS helped to draft the manuscript. TM and YK participated in its design and coordination.